Author’s response to reviews

Title: Using an onset-anchored Bayesian hierarchical model to improve predictions for amyotrophic lateral sclerosis disease progression

Authors:

Alex Karanevich (akaranevich@kumc.edu)
Jeffrey Statland (jstatland@kumc.edu)
Byron Gajewski (bgajewski@kumc.edu)
Jianghua He (jhe@kumc.edu)

Version: 2 Date: 20 Jan 2018

Author’s response to reviews:

We received ethical approval from the University of Kansas Medical Center’s Human Research Protection Program. This was added to our ethical statement as "This project was deemed not to constitute human subjects research and the need for ethical approval for this work was waived by the University of Kansas Medical Center’s Human Research Protection Program." For your convenience, this has been changed to blue text so that it stands out.